

Transfer line balancing problem: A comprehensive review, classification, and research avenues

Pedram Beldar, Masood Fathi, Amir Nourmohammadi, Xavier Delorme, Olga

Battaïa, Alexandre Dolgui

▶ To cite this version:

Pedram Beldar, Masood Fathi, Amir Nourmohammadi, Xavier Delorme, Olga Battaïa, et al.. Transfer line balancing problem: A comprehensive review, classification, and research avenues. Computers & Industrial Engineering, 2025, pp.110913. 10.1016/j.cie.2025.110913 . hal-04925450

HAL Id: hal-04925450 https://hal.science/hal-04925450v1

Submitted on 20 Feb 2025

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

Contents lists available at ScienceDirect



Computers & Industrial Engineering



Computers & Computers & Industrial Engineering Workshow of the Computer of the Comput

Transfer line balancing problem: A comprehensive review, classification, and research avenues^{*}



Pedram Beldar ^a, Masood Fathi ^{a,b}, Amir Nourmohammadi ^a, Xavier Delorme ^b, Olga Battaïa ^d, Alexandre Dolgui ^b

^a Division of Intelligent Production Systems, School of Engineering Science, University of Skövde, P.O. Box 408, SE 541 28 Skövde, Sweden

^b Division of Industrial Engineering and Management, Uppsala University, PO Box 534, Uppsala 75121, Sweden

^c Mines Saint-Etienne, Univ Clermont Auvergne, INP Clermont Auvergne, CNRS, UMR 6158 LIMOS, F - 42023 Saint-Etienne, France

^d Kedge Business School, 680 cours de la Libération, 33405, Talence, France

e IMT Atlantique, LS2N - UMR CNRS 6004, F-44307 Nantes, France

ARTICLE INFO

Keywords: Transfer line Machining line Line balancing Line design Systematic review

ABSTRACT

The Transfer Line Balancing Problem (TLBP) is characterized as the challenge of optimally distributing tasks across various workstations in an automated machining line to ensure its maximum efficiency. This problem holds pivotal importance for industries reliant on high-volume production, such as the automotive and aerospace sectors, where it directly influences the overall productivity and cost efficiency of the manufacturing process. TLBP has been studied for over two decades, and many problem variants and solution approaches have been devised to address real-world challenges. Despite the long history of the topic, no review study exists to shed light on its past, current, and future developments. This study conducted a systematic literature review on TLBP to identify and address the research gaps, focusing on classifying existing studies. A tuple notation classification framework has been introduced to organize TLBP research based on system configuration, problem characteristics, and optimization objectives. This framework offers a structured overview of the field, clarifying the current state of research and highlighting prospective research pathways. Consequently, this review study establishes itself as a foundational guide for academics and industry professionals interested in TLBP studies.

1. Introduction

In today's ever-changing industrial landscape, intensified by globalization, manufacturers are dealing with an environment filled with intense competition (Lahrichi, Grangeon et al., 2021). They are challenged by fluctuating consumer demands, frequent updates in product designs, and continuous shifts in manufacturing technologies. To effectively tackle these challenges, a thorough revision and enhancement of their production systems are essential. Machining is still a major technology in modern production systems. In this context, the optimal design of automatic machining lines, commonly known as transfer lines, becomes particularly significant (Lahrichi, Deroussi et al., 2021). These transfer lines are instrumental in ensuring high efficiency, substantial output, and economical production, thereby playing a crucial role in maintaining manufacturers' competitive strength and their adaptability to market dynamics (Liu et al., 2016).

Transfer lines, entailing a sequence of workstations and an advanced transport mechanism, ensure a seamless flow of parts through various machining operations (Essafi, Delorme, Dolgui, Guschinskaya, 2010). Noted for their capacity to produce high volumes of identical or similar products (Borisovsky et al., 2012b), transfer lines are indispensable in high-demand sectors like automotive and aerospace manufacturing. The key advantages of transfer lines are their ability to guarantee high productivity, product quality, operational efficiency, and cost-effectiveness, achieved through consistent operations conducted at each workstation in every cycle (Osman & Baki, 2018). Despite the benefits of transfer lines, their efficiency significantly relies on successfully addressing several challenges at different decision levels that require

https://doi.org/10.1016/j.cie.2025.110913

Available online 29 January 2025

0360-8352/© 2025 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

¹The first three authors would like to acknowledge funding from the Knowledge Foundation (KKS) and Sweden's Innovation Agency through the ACCURATE 4.0 (grant agreement No. 20200181) and PREFER projects, respectively.

^{*} Corresponding author at: Division of Intelligent Production Systems, School of Engineering Science, University of Skövde, P.O. Box 408, SE 541 28 Skövde, Sweden.

E-mail addresses: pedram.beldar@his.se (P. Beldar), masood.fathi@his.se (M. Fathi), amir.nourmohammadi@his.se (A. Nourmohammadi), delorme@emse.fr (X. Delorme), olga.battaia@kedgebs.com (O. Battaïa), alexandre.dolgui@imt-atlantique.fr (A. Dolgui).

careful optimization. One of these challenges is line balancing, which comprises assigning tasks to workstations while adhering to a range of constraints to optimize one or several criteria, such as the number of machines, cycle time, or/and the total cost of the line, to name a few (Battaïa et al., 2024; Liu et al., 2021). While line balancing has been a common practice in manual assembly lines, its importance extends to transfer lines, with unique considerations due to specific features of machining technologies. The process of line balancing in transfer lines is complex and multifaceted, encompassing steps like selecting operations, creating a logical layout, and designing the necessary equipment for each station (Dolgui et al., 2006b) with the possibility of parallel processing of operations at stations. This led to the introduction of a specific problem known as the Transfer Line Balancing Problem (TLBP) in the literature (Dolgui et al., 1999).

Despite the multifaceted nature of TLBP, which has captured considerable attention for over two decades, no studies can be found providing a comprehensive classification of existing research based on varied criteria such as system configuration, problem specifications, and optimization objectives. Although there are recent review papers on line balancing, the majority of them focus on assembly lines (e.g., Battaïa & Dolgui, 2022; Boysen et al., 2022) or disassembly lines (e.g., Deniz & Ozcelik, 2019; Eren Özceylan & Gupta, 2019). Some studies, such as Battaïa and Dolgui (2022), review machining line balancing, but their purpose was to identify hybrid approaches to line balancing, integrating other optimization problems such as process planning, workforce planning, and/or resource scheduling. Therefore, they include only a few studies on TLBP, underscoring the need for a comprehensive classification that spans various aspects of TLBP research.

This gap is significant due to the complexity of TLBP, which is influenced by strategic, tactical, and operational decision-making, as well as by the diversity of transfer lines. Such a classification is not merely a matter of academic interest; it serves as a road map for manufacturers navigating the challenges of TLBP. A well-defined classification would provide clarity in problem formulation. Researchers would gain a structured understanding of the various aspects of TLBP, helping them in defining problems more precisely. Each category within the classification may necessitate specific optimization techniques. Researchers could adjust their approaches to address the unique challenges posed by different types of TLBPs. Professionals in the field would benefit from precise solutions designed to address the specific challenges within their industry. This would result in more efficient and cost-effective production systems. A comprehensive classification would encourage inclusive research, allowing for a deeper understanding of the correlations among strategic, tactical, and operational decisions. This could lead to innovative solutions that bridge the gaps between these levels.

Given the absence of thorough classification in TLBP literature, this study endeavors to bridge the aforementioned gaps by:

- 1. conducting a systematic literature review (SLR) on TLBP to provide a comprehensive knowledge of the subject.
- introducing a tuple notation classification framework that considers different system configurations, problem characteristics, and optimization objectives, providing a structured procedure to classify various TLBP scenarios.
- 3. presenting the state-of-the-art literature using customized tuple notations to enhance the clarity and precision in the field of TLBP.
- providing insights into future research directions for TLBP, paving the way for more exploration and development in this area.

The rest of this article is divided into the following sections: Section 2 will provide an overview of the TLBP definitions, forming the foundation for the classification scheme that is unveiled in Section 4. In Section 3, the research methodology employed for the literature review is discussed. Section 4 presents the proposed classification scheme.

Section 5 utilizes the classification approach to arrange the body of current literature. Section 6 analyzes the findings from the literature classifications. Section 7 outlines potential future research directions. Finally, Section 8 presents the conclusions drawn from this review.

2. Transfer line balancing problem

Line balancing has long been a fundamental problem in the operational management of manufacturing systems, particularly concerning the efficient allocation of tasks among workstations along a production line (Essafi, Delorme, Dolgui, Guschinskaya, 2010). Historically, the problem first received academic attention in the realm of assembly lines (Salveson, 1955). The primary goal of assembly line balancing (ALBP) is to assign tasks to stations such that one or more objectives (based on time, cost, and efficiency measures) are optimized. This problem has been thoroughly explored over the past decades, with numerous methodologies developed to address its inherent NP-hard complexity (see Boysen et al., 2022). Despite the extensive research on ALBP, the distinct nature of machining processes led to the emergence of the TLBP, which focuses on production lines that involve machining operations (Dolgui & Ihnatsenka, 2009b). The need to consider unique features of machining posed a new challenge that the solutions developed for ALBP could not directly address (Dolgui et al., 2010). The key differences are as follows (Dolgui et al., 2020; Essafi et al., 2009; Essafi, Delorme, Dolgui, Guschinskaya, 2010):

- Automation: TLBP are characterized by a high degree of automation involving sophisticated machinery and automated material handling systems, whereas ALBP typically includes more manual operations.
- **Parallel processing**: TLBP enables simultaneous operations through multi-spindle heads or parallel machines, enhancing productivity. In contrast, ALBP typically performs tasks sequentially.
- Equipment design (selection): In TLBP, equipment design (selection) is essential and must be addressed alongside line balancing, as machinery and tool choices directly influence line configuration and performance. Task times are often unknown prior to optimization, with only machining operation parameters known (such as working stroke and feed rate). In ALBP, equipment design (selection) is important but generally less central to the balancing process than in TLBP.
- · Specific constraints: Numerous specific constraints are associated with the unique characteristics of machining processes. For instance, zoning constraints are more complex, concerning not only the stations but also the designed (selected) equipment. In TLBP, different machining centers have varying features that directly influence task grouping or separation, adding a layer of complexity that is not typically present in ALBP. Unlike ALBP, where tasks are typically performed sequentially, TLBP often involves parallel processing using multi-spindle heads or multiple machines at a single station. This capability introduces additional considerations for zoning constraints, as tasks assigned to the same station must be compatible for simultaneous execution without causing operational conflicts. The choice of machinery and tools directly influences the feasibility and efficiency of task groupings, as different equipment configurations may enable or restrict certain zoning options. For example, selecting a multispindle machine may allow for positive zoning of compatible tasks, while choosing specialized tools might necessitate negative zoning for certain operations.
- Adjustability of cutting speed (feed rate): In machining, cutting speed (feed rate) can be adjusted to modify task durations, allowing adjustments for specific equipment and tasks. Thus, transfer line balancing combines balancing, process, and equipment design optimization.

The TLBP concept was first introduced by Dolgui et al. (1999) for mass production of a single type of product (i.e., a product model), marking a shift in line balancing research from assembly to machining. In a typical TLBP, the line is paced and serial, consisting of stations connected by automated material handling systems (Dolgui et al., 2006d). Parts move automatically from one station to the next, maintaining an uninterrupted flow (Osman & Baki, 2014). Each station is equipped with specialized tools that perform machining operations using single or multiple spindle heads on the part.

In the literature, the emergence of various transfer lines has promoted the development of specialized TLBPs, each type of transfer line designed to address the changing demands of modern manufacturing sectors. Traditional Dedicated Transfer Lines (DTLs) are designed for high-volume production of a single model (Guschinskaya & Dolgui, 2008), optimizing productivity and minimizing cost per unit. However, as market demands shifted, more adaptable systems emerged. Flexible Transfer Lines (FTLs) were introduced to handle small sets of various models within a large product family. These systems employ flexible machines, typically machining centers with Computer Numerical Control (CNC), and advanced material handling systems (Essafi et al., 2010b). FTLs, however, are costly and less productive. Responding to the growing need for both large-volume production and market flexibility, Reconfigurable Transfer Lines (RTLs) have gained prominence (Borisovsky et al., 2014). RTLs offer a balanced solution between the high production rates of DTLs and the flexibility of FTLs. RTLs use modular components for quick reconfiguration, allowing for efficient handling of a single model or a small family of models (i.e., customized flexibility). RTLs are also characterized by their scalability, allowing for modification of the production volume.

3. Research methodology

To present a systematic review of TLBP, this study followed the PRISMA protocol (Moher et al., 2009), comprising (i) establishing specific research questions, (ii) identifying reputable information sources, (iii) organizing the search approach, (iv) defining criteria for selecting relevant papers, (v) extracting important data methodically, and (vi) examining and discussing the findings thoroughly.

3.1. Research inquiries

The design and configuration of transfer lines exhibit different types, each characterized by distinct characteristics and objectives. This investigation delves into the landscape of TLBP, aiming to uncover the critical questions and insights that define their optimization. To guide this exploration, the following research questions (RQ) are proposed:

RQ1: How can a classification framework be developed to categorize TLBP literature effectively?

RQ2: What current trends and thematic focuses have shaped the TLBP field?

RQ3: What are the future research directions in TLBP?

3.2. Data sources and search approach

This study conducted an exhaustive exploration of prominent peerreviewed scientific databases, Scopus, Web of Science, and Science Direct, to systematically identify and evaluate the most relevant research papers. The search stream was constructed by considering two components linked by the "AND" operator. The primary search component involved the consideration of three keywords, "manufacturing system", "transfer line" and "machining line" with a specific focus on retrieving articles with these terms in their titles. Given these three keywords are broad terms that can span various fields, the inclusion of "line balancing" and "line design" keywords in the "title, abstract, and keywords" was employed as the second search component. This approach aimed to refine the search results, ensuring the retrieval of the most pertinent articles (see Fig. 1).

3.3. Inclusion and exclusion guidelines

Comprehensive inclusion and exclusion criteria were employed to either include or exclude a paper from the list. Concerning exclusion criteria, three distinct conditions were implemented: (a) papers not written in English, (b) the title and abstract review, and (c) a full-text review. No restrictions were placed on publication years, allowing for a comprehensive coverage of the field from its inception to the present. On the other hand, the inclusion criteria were established based on the exclusive consideration of journal and conference papers.

3.4. Paper extraction

As illustrated in Fig. 1, the initial search yielded a total of 250 research papers. Subsequently, 84 duplicate papers and book chapters were excluded from the list. Furthermore, 9 papers that were not written in English were eliminated. During the meticulous "title and abstract" review, 66 papers were also removed from the list since they addressed topics that lay outside the scope of manufacturing systems. Additionally, 25 papers were excluded upon conducting a full-text review of the remaining papers.

3.5. Snowball search

To ensure comprehensive coverage of the TLBP literature, a snowball search approach was employed to supplement the initial database search (Wohlin, 2014). This approach involved two complementary strategies. First, a "backward search" was conducted by examining the reference lists of key articles identified in the initial search. This allowed for tracing back to foundational studies and identifying seminal works frequently cited in the field. Second, a "forward search" was performed to identify articles that cited these key references, ensuring the inclusion of more recent studies building upon the foundational works. Notably, some relevant papers, including those containing terms such as "rotary" did not appear in the initial search results. The snowball search method effectively addressed this gap by identifying and incorporating these studies. The combined use of forward and backward search methods yielded 18 additional papers that met the inclusion criteria. These papers were subsequently incorporated into the review, bringing the total number of papers analyzed in this study to 84. By employing this rigorous and systematic approach, the review ensures comprehensive coverage of the literature in the field of TLBP, capturing both foundational works and recent advancements.

A comprehensive dataset of 84 papers was established through the systematic review process, forming the basis for a structured approach to analyzing and categorizing this literature. In the following section, a novel classification framework is introduced to organize and analyze the diverse body of TLBP research. This framework aids in understanding the current state of the field and provides a foundation for identifying research trends and potential gaps in the literature.

4. The classification framework

Drawing on the systematically collected literature, this section addresses the first research question by presenting a comprehensive classification framework for TLBP research. Numerous proficient and thoughtfully designed frameworks have been presented across diverse academic domains. These frameworks serve the purpose of addressing the inherent challenges within their respective fields of study and contribute to the clarity of problem formulation. Inspired by the renowned and effective classification frameworks pioneered by Graham et al. (1979) for machine scheduling, Brucker et al. (1999) for project scheduling, and Battaïa and Dolgui (2013) and Boysen et al. (2007) for assembly line balancing problems, a novel classification approach is introduced for the TLBP based on tuple-notation. This approach seeks to provide a structured and systematic framework for categorizing



Fig. 1. PRISMA Method: the process of collecting papers.

TLBPs, drawing upon the successes observed in related fields. The framework aims to enhance the understanding of TLBPs and pave the way for more targeted research efforts in this critical domain of manufacturing optimization. Therefore, each TLBP is encapsulated by three fundamental components, symbolized as $\alpha \mid \beta \mid \gamma$. First, α serves as the identifier of the TLBP system configuration, helping to classify it within the broader landscape of manufacturing challenges. Second, β outlines the specific characteristics and attributes taken into account when formulating the problem, creating a detailed profile of the TLBP. Finally, γ signifies the objective function, clarifying the ultimate goal and optimization criterion guiding the problem-solving process.

In the following subsections, the details of this tuple-notation classification scheme are discussed, with its key components highlighted and its potential applications in the context of TLBPs illustrated.

4.1. Classification of TLBP according to line type (α)

In this section, the classification of TLBP is considered in order to distinguish between various systems using a three-part format: $\alpha_1\alpha_2(+\alpha_3)$. Here, α_1 represents the system configuration (S for serial, R for rotary, PS for parallel-serial, C for cellular), α_2 denotes the number of models considered during the design stage (0 for existing line, 1 for single model, n for mixed models, nb for multi-model), and the optional α_3 indicates reconfigurability, showing if additional models can be integrated later. For instance, a DTL could be *S*1 or *R*1, a FTL could be *Sn* or *Cn*, and a RTL could be *PS*1 + 0, *PS*1 + *n*, *PSn* + 0, or *PSn* + *n*. Also, a paper dealing with the reconfiguration of an existing line could be *S*0 + 1 or *PS*0 + 1. Therefore, a serial line for a single model is labeled as S1, while a rotary line for a family of models, considering several different models at the design stage, is



Fig. 3. Classification of problem specifications.

Rn. A parallel–serial line initially designed for one model but allowing future unknown reconfiguration is PS1+0, and one considering multiple future models is PS1+n. Additionally, a line initially designed for a single model, now being reconfigured to integrate one additional model is denoted S0+1 or PS0+1. Fig. 2 illustrates the classification for line types considered in TLBP, showing how configurations, number of models, and reconfigurability are represented in the notation $\alpha_1\alpha_2(+\alpha_3)$.

4.2. The problem specifications (β)

In this section, key attributes of TLBP are systematically organized into three primary categories: equipment-related characteristics, timerelated characteristics, and operation assignment characteristics. This organizational approach establishes a solid base for a detailed understanding of TLBP, making the investigation of its complex elements more straightforward. The specifics of this β classification will be discussed in the subsequent sections. Fig. 3 provides an overview of the problem specifications, illustrating the classification of key attributes into equipment-related, time-related, and operation assignment categories.

4.2.1. Equipment-related characteristics

In the existing body of literature, considerable attention has been devoted to incorporating various equipment-related properties associated with machines in workstations. By incorporating these equipmentrelated elements, researchers aim to create TLBP scenarios that closely reflect practical industrial settings.

 $\beta = A^{io}$. In machining operations, the precise tool-task relationship is critically essential. For example, using a specific tool for a designated

task reflects the careful selection and sequencing of operations. This precise assignment of tools to their respective tasks is a key element in modern machining. It ensures that each operation is performed with the most suitable tool, underscoring the detailed planning and efficiency necessary in these processes.

 $\beta = A^f$. Allocation of specific sides of a workpiece to dedicated fixtures is also a crucial aspect considered in machining operations. This process ensures that operations are accurately performed on the specified surfaces and features of the workpiece, highlighting the importance of precision in fixture allocation for successful machining outcomes.

 $\beta = A^{tu}$. Turret assignment involves the allocation of individual tasks or operations to specific turrets. This assignment process plays a pivotal role in optimizing the machining operations across various workstations, ensuring efficiency, precision, and effective task management. Turrets in the machining process serve as a vital component for accessing different sides of workpieces at various working positions. These turrets facilitate both sequential and parallel activation of spindle heads or machining modules to carry out operations on the parts. They enable the machinery to work efficiently on the components by allowing horizontal and vertical spindle heads, whether separate or mounted in turrets, to access different sides of the workpiece. These turrets, which can hold multiple machining tools, are instrumental in executing operations in a specific sequence.

 $\beta = Ac^C$. Accessibility constraints in the context of part positioning are crucial considerations. In specific orientations, certain sides of a part may become inaccessible, necessitating repositioning to facilitate the required operations (Essafi et al., 2009).

 $\beta = To^{ma}$. A tool magazine is a storage system predominantly associated with CNC machining centers, containing various types of

cutting tools needed for different machining operations. These tools are loaded onto the CNC machine while adhering to the magazine's capacity constraint (Osman & Baki, 2014). In a CNC machining center, when one operation is completed with a specific cutting tool, and the next operation requires a different tool, the machine's spindle returns to the tool magazine to retrieve and load the required tool for the next step in the machining process (Bhale et al., 2014). Essentially, a tool magazine in a CNC machining center ensures that the right tools are readily available for efficient and seamless machining operations.

 $\beta = Bu^{al}$. Buffer allocation between stations is a key factor in improving the independence of sequential workstations within a process under random disturbances (Shao et al., 2020). This strategy primarily aims to allow these workstations to function more independently from each other, thereby streamlining the overall workflow and improving operational efficiency.

 $\beta = SH^{mono}$: A mono-spindle head machine can sequentially use different tools and rotate the part to perform various operations, moving to each position one at a time in a machining operation (Essafi et al., 2010a).

 $\beta = SH^{multi}$: A multi-spindle head machine can simultaneously use multiple tools to perform various operations in parallel (Dolgui & Ihnatsenka, 2009c).

4.2.2. Time-related characteristics

 $\beta = Sd^{st}$. The focus on sequence-dependent setup times highlights that the time needed for two consecutive operations in a manufacturing process is affected by the particular sequence in which they are performed. This is because factors such as tool changes and table or part rotation introduce variability in setup times (Essafi et al., 2010a).

 $\beta = TC^{ti}$. The consideration of tool change time is essential, especially when a different tool is required to perform another operation (Osman & Baki, 2014).

 $\beta = TuR^{ii}$. The time taken for the turret to rotate affects the overall efficiency and flow of operations within the machining cycle (Afrin et al., 2016).

 $\beta = Re^{ti}$. Considering the time needed for refixturing or reloading is a vital part of the process.

 $\beta = LU^{ti}$. Addressing the loading and unloading time of a part on a machine is a crucial aspect of the process.

 $\beta = TaR^{ti}$. The consideration of table rotation time is crucial in the machining process. As the table rotates, it relocates the workpiece to various working positions, each designated for specific operations (Afrin et al., 2016).

 $\beta = Tr^{ii}$. Addressing the transportation time between stations is a key aspect of the process. It involves the time taken to move materials or parts from one station to another.

 $\beta = OC^{ii}$. The time required for an orientation change is an essential aspect of the process. This occurs when an operation needs to be performed on a different surface of the workpiece (Osman & Baki, 2014).

4.2.3. Operation assignment characteristics

 $\beta = C^{prec}$. Precedence constraints refer to the established order of tasks or operations. This order dictates a partial sequence for all tasks and operations to be executed (Dolgui et al., 2006a). These constraints define a set of feasible task/operation sequences (Essafi, Delorme, Dolgui, Guschinskaya, 2010), ensuring that certain tasks/operations must be completed before others can begin, thereby dictating the acceptable order in which tasks/operations are executed. For systems such as multi-spindle heads with parallel execution of operations, an extended definition of precedence constraints is applied. Specifically, if task *a* precedes task *b*, this means that task *a* should be executed before task *b* or simultaneously with task *b*, i.e., in parallel. This flexibility accommodates scenarios where parallelism is inherent to the operational process.

 $\beta = C^{inc}$. Inclusion constraints, which specify the grouping of tasks/operations that must share a workstation or executed with the same equipment (e.g., multi-spindle head), are often driven by essential machining tolerance considerations (Guschinskaya & Dolgui, 2009).

 $\beta = C^{exc}$. Exclusion constraints are rules preventing particular sets of operations from being allocated to the same workstation or piece of equipment due to technological incompatibility (Guschinskaya & Dolgui, 2009).

 $\beta = C^{tec}$. Technological constraints dictate that certain task can only be executed on specific machine due to compatibility requirements (Nourmohammadi et al., 2024).

 $\beta = C^{acc}$. Task-machine accessibility constraints specify which tasks are feasible on particular machines. This constraint ensures that a task can only be assigned to machines that are capable of performing it based on compatibility and accessibility requirements (Nourmohammadi et al., 2024).

 β = *Seq.* Sequencing addresses the tasks/operations sequences, which refer to determining the order in which tasks or operations should be performed at various workstations or positions within the line (Lahrichi, Grangeon et al., 2021).

 β = *Sch*. Scheduling refers to determining the specific start and finish times for each task to be performed at various workstations.

4.3. Objectives (γ)

The optimization of TLBP is motivated by the aim of achieving specific objectives that act as metrics for assessing solutions. These objectives can be grouped into three primary categories: time-related objectives (operational level), cost-related objectives (tactical or strategic level), and efficiency-related objectives (tactical level). In some cases, a research paper can often include multiple objectives. Fig. 4 categorizes the objectives for optimizing TLBP into time-related, cost-related, and efficiency-related metrics, highlighting the main objectives used to evaluate solutions.

4.3.1. Time-related objectives

 $\gamma = NonP^{ti}$. The concept of non-productive time (NPT), encompasses various activities such as tool change time, reloading time, transportation time, reposition time, refixturing time, and any other tasks that do not directly contribute to the actual production of a workpiece. Minimizing NPT enhances transfer line productivity by shortening the production lead time (Osman & Baki, 2018).

 $\gamma = Cyc^{ti}$. The cycle time is determined by the load on the bottleneck workstation, which represents the longest time required among all workstations (Telemeci & Azizoğlu, 2024). For manufacturers heavily invested in transfer lines, reducing the cycle time is crucial for maximizing production rates. Since the transfers are automated and synchronized, minimizing cycle time also reduces the flow time, which is the duration each product unit spends in the system.

 $\gamma = SI$. The goal is to minimize the smoothness index by evenly distributing processing time and keeping cycle time variations at each workstation as close to the line cycle time as possible, thus promoting a uniform workload distribution throughout all workstations (He et al., 2020).

 $\gamma = Pro$. The goal is to minimize the probability of workstation cycle time exceeding the line cycle time (He et al., 2020).

4.3.2. Cost-related objectives

 $\gamma = Li^{co}$. Consideration is given to the comprehensive cost of designing and equipping the production line. This includes the strategic minimization of investment required by reducing the total number of stations as well as evaluating expenses associated with installing various types of equipment such as turrets, machining modules, and spindle heads (Battaïa et al., 2020; Guschinskaya & Dolgui, 2006).



Fig. 4. Classification of objectives for TLBP.

 $\gamma = En^{co}$. Energy related cost is considered. This can include Time-Of-Use energy prices, power peak or total energy consumption minimization (Delorme & Gianessi, 2024).

 $\gamma = Man^{co}$. Manufacturing cost optimization is targeted, with a focus on minimizing the annual costs of machinery and tooling (Osman & Baki, 2018).

 $\gamma = To^{co}$. All costs are considered, including design and manufacturing costs. Machines and buffers are taken into account within the concept of total cost.

 $\gamma = Rec^{co}$. Reconfiguration costs are taken into account, which include costs related to equipment, labor, and order loss (Liu et al., 2021).

 $\gamma = NM$. The aim is to minimize the total number of machines installed on the workstations.

4.3.3. Efficiency-related objectives

 $\gamma = PR$. The focus is placed on maximizing the production rate, which serves as a measure of the throughput of the production line (Shao et al., 2020).

 $\gamma = LBR$. The objective includes maximizing the line balancing rate (Liu et al., 2016). It is a critical parameter that is maximized when the total workload is equitably allocated across all workstations in proportion to the number of machines and the overall number of workstations within the system.

 γ = *CE*. The study considers the minimization of the carbon emissions (Afrin et al., 2016).

 γ = *Scal*. The objective is to maximize the scalability of the production line, i.e., its ability to adjust production capacity (Cerqueus & Delorme, 2023).

 $\gamma = Se^{le}$. It refers to maximizing the service level (Delorme et al., 2023).

 $\gamma = SR$. The objective is to maximize the stability radius (Pirogov et al., 2021). The stability radius measures the robustness of a feasible solution in the face of uncertainty. It is defined as the maximum amount of processing time increase that a solution can withstand while remaining feasible, regardless of how this extra processing time is distributed among uncertain tasks. The stability radius is computed by determining the radius of the largest closed ball, called the stability ball, that contains all feasible solutions within a certain distance from the nominal solution.

4.4. Examples of TLBP classification

This section provides examples of how the proposed framework can be applied to classify TLBP across a range of industrial scenarios. The framework's versatility is demonstrated through three representative cases, each highlighting a unique production configuration and optimization objective.

In the first scenario, a manufacturer aims to design a reconfigurable transfer line for a single product model, with the flexibility to accommodate new models in the future. This problem can be expressed as $PS1 + 0 | C^{prec}, Ac^C, SH^{mono} | NM$ where the classification captures the essential features of the system. Specifically, the designation PS1 + 0 refers to a parallel-serial line configurability. Constraints related to precedence (C^{prec}) and accessibility (Ac^C) are incorporated into the design. The use of mono-spindle heads (SH^{mono}) ensures efficient machining processes, while the primary objective of the optimization (NM) is to minimize the number of machines required.

A second scenario involves a flexible transfer line intended for mixed-model production. This system is represented as $Sn \mid A^{to}, To^{ma}$, $Sd^{st}, SH^{mono} \mid NonP^{ti}$ indicating a serial line (Sn) capable of handling mixed models. The classification accounts for tool assignment (A^{to}) , tool magazine capacity (To^{ma}) , and sequence-dependent setup times (Sd^{st}) . The production setup employs mono-spindle heads (SH^{mono}) , while the optimization goal $(NonP^{ti})$ focuses on minimizing non-productive time, thereby enhancing the system's operational efficiency.

The final example considers a rotary transfer line designed for single-model production, which can be characterized as $R1 | A^{tu}, TaR^{ti}$, $TuR^{ti}, SH^{multi} | Li^{co}, CE$. Here, R1 signifies a rotary line configuration tailored to single-model manufacturing. The framework incorporates critical factors such as turret assignment (A^{tu}), table rotation time (TaR^{ti}), and turret rotation time (TuR^{ti}), reflecting the complexity of rotary systems. Multi-spindle heads (SH^{multi}) are employed to maximize productivity, while the optimization objectives include minimizing both line cost (Li^{co}) and carbon emissions (CE).

These examples underscore the flexibility and comprehensiveness of the TLBP classification framework, demonstrating its ability to represent diverse production scenarios systematically and to support targeted optimization strategies.

Table 1

Abbreviations for the approaches and decision-making levels.					
Exact methods	Decomposition approach (D), Graph approach (G), Branch and Bound (B&B), Branch and Cut (B&C), Polynomial algorithm (PA), Dynamic programming (DP), Set partitioning (SP), Reduction method (RM), Constraint generation (CG)				
Metaheuristics	Greedy randomized adaptive search procedure (GRASP), Non-Dominated Sorting Genetic Algorithm II (NSGA-II), Genetic Algorithm (GA), Ant colony optimization (ACO), Cuckoo search (CS), Multi-Objective Particle Swarm Optimization (MOPSO), Artificial Bee Colony (ABC), Tabu search (TS)				
Other approaches	Mathematical model (MM), Heuristic (H), Simulation (Sim), Matheuristic (MH), Combinatorial approach (CA)				
Decision-making level	Strategic (ST), Tactical (TA), Operational (OP)				



Fig. 5. Frequency of different system configurations for three types of transfer lines.

5. The literature classification

This section also contributes to addressing the first research question by systematically classifying the TLBP studies following the framework detailed in Section 4. For the sake of completeness, the solution approach used and the targeted decision-making level are also reported for each study. In general, the TLBPs encompass strategic, tactical, and operational decision-making levels, each critical to the production line's performance. Strategically, decisions dictate the line's overall configuration, impacting long-term efficiency and adaptability. Tactically, the focus shifts to the assignment of tools and equipment, which is crucial for optimizing throughput and flexibility. Operationally, tasks are allocated to workstations, directly influencing day-to-day efficiency and productivity. These intertwined layers of decision-making shape the line's ability to meet current production demands while retaining the agility to adjust to future changes, underlining the multifaceted nature of TLBP.

For clarity and brevity, abbreviations are proposed for the distinct solution approaches and decision-making levels, as shown in Table 1. Following this, Table 2 presents a comprehensive chronological analysis of each publication, incorporating notations, decision-making levels, and solution approaches employed. The papers are arranged in ascending order by publication year to illustrate the evolution of TLBP research over time.

6. Analysis of classification findings

This section contributes to addressing the second research question. It provides an overview of the main findings according to the literature classification. Fig. 5 shows the frequency of different system configurations for three types of transfer lines: DTL, FTL, and RTL. This distribution indicates that the DTL configuration 'S1' is the most discussed in the literature, suggesting that it may be the most common or established type of system configuration. The RTL configuration 'PS1+0' is also frequently mentioned, which may indicate a growing preference due to their efficiency and flexibility balance. Other configurations like 'R1', 'C1+0' and 'PS0+n' are less commonly discussed. FTL configurations, although essential for flexibility in production, are less frequently mentioned than DTLs and RTLs. Similarly, we found no paper considering configuration 'Cn', which could mean that authors focus on other problems than line balancing when they study such systems.

Fig. 6 presents a heatmap showcasing the frequency of various objectives concerning different line types considered in TLBP, with a striking predominance of cost-related objectives. This focus on cost is significant, constituting 78.49% of the literature, which underscores its critical role in shaping TLBP strategies, particularly within DTL and RTL, where it is most pronounced. The 'Line cost' objective, as shown in Fig. 7, is particularly notable, appearing 53 times, underscoring its significance in the context of TLBP considerations. The emphasis on 'Number of machines' with frequencies of 13, further reinforces the priority given to cost optimization in equipment and resource allocation. The lower frequency of time-related objectives such as 'Cycle time' (with a frequency of 8) and 'Non-productive time' (with a frequency of 3), in conjunction with their smaller representation in the heatmap (13.98%), suggests that these factors are less prioritized. However, they are still important as they directly impact throughput and efficiency. In the data depicted by the provided plots, the 'Production rate' and 'Line balancing rate' objectives appear 3 and 1 times, respectively, suggesting their relevance to enhancing operational efficiency within TLBP strategies, albeit less emphasized compared to cost-related considerations.

There is limited literature focusing on sustainability aspects such as energy efficiency, emissions reduction, and overall eco-performance in the context of TLBP. The predominant focus of the existing studies is on cost and time optimization, with minimal direct attention given to the environmental impact of TLBP strategies. This gap in the literature highlights a crucial area for future research, underscoring the need to integrate sustainability considerations into TLBP analysis and decision-making.

According to Fig. 8, most research in the field of TLBP focuses on single-objective (SO) optimization. Based on the chart, it can be observed that 45 studies for the DTL are centered around SO, whereas just three studies address multi-objective (MO) problems. Other line types also exhibit this tendency, with RTL displaying a lower percentage of studies (26 SO and seven MO studies, respectively). FTL has the fewest studies, with three papers focusing on SO and no paper on MO optimization. The strong tendency towards SO studies suggests that most research simplifies the problem to a single dimension. A very small fraction of the studies are MO, which implies considering two or more objectives simultaneously. This is the most complex approach

P. Beldar	• et al.
-----------	----------

Table 2

Authors	Line type	Notation	Decision-making level	Approach
Dolgui et al. (1999)	DTL.	S1 Cprec, Cinc, Cexc, SH ^{multi} Li ^{co}	ST. OP	MM. G
Dolgui, Guschinsky, Levin	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	MM, D
(2000)		• • • • •		
Dolgui, Guschinsky, Levin et al. (2000)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	MM, G
Bratcu et al. (2003)	DTL	S1 C ^{prec} , C ^{inc} , C ^{exc} , A ^{lo} , SH ^{mulli} Li ^{co}	ST, OP	MM, B&B
Dolgui et al. (2003)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	MM, H
Dolgui et al. (2004)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	MM, H
Bratcu et al. (2005)	FTL	$Sn \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	MM
Dolgui, Finel et al. (2005)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	Н
(2005)	DIL	$S1 \mid C^{pee}, C^{out}, C^{eee}, SH^{outain} \mid Li^{eo}$	S1, OP	н
Dolgui and Ihnatsenka (2005)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	B&B
Battaïa et al. (2005)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	Н
Belmokhtar et al. (2006)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{mulli} \mid Li^{co}$	ST, OP	MM
Masood (2006)	RTL	$PS1 + 0 C^{prec}, SH^{mono} Cyc^{ll}$	ST, OP	Sim
Dolgui et al. (2006b)	DTL	SI C ^{prec} , C ^{inc} , C ^{exc} , SH ^{mult} Li ^{co}	ST, OP	MM, H
Dolgtil et al. (2006c)	DIL	$S1 \mid Cprec, Cinc, Cexc, SH main \mid Lico$	SI, OP	MM, D
(2006)	DIL		31, OP	ми, н
Dolgui et al. (2006d)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{mull} \mid Li^{co}$	ST, OP	G
Dolgui et al. (2006a)	DTL	SI C ^{prec} , C ^{inc} , C ^{exc} , SH ^{multi} Li ^{co}	ST, OP	D
Guschinskaya et al. (2007)	DTL	S1 C ^{prec} , C ^{inc} , C ^{exc} , SH ^{mult} Li ^{co}	ST, OP	RM
(2007)	DIL	$S1 \mid C^{pee}, C^{out}, C^{exe}, SH^{main} \mid Li^{out}$	S1, OP	GRASP
Dolgui, Guschinsky, Levin et al. (2008)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{mulli} \mid Li^{co}$	ST, OP	G, MM, H
Dolgui, Guschinsky, Levin et al. (2008)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	G
Finel et al. (2008)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	Н
Guschinskaya et al. (2008)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, LU^{ti}, SH^{multi} \mid Li^{co}$	ST, OP	Н
Dolgui, Guschinskaya, Guschinsky et al. (2009)	DTL	$R1 \mid C^{prec}, C^{inc}, C^{exc}, LU^{tl}, SH^{multi} \mid Li^{co}$	ST, OP	MM
Dolgui and Ihnatsenka (2009a)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	SP
Dolgui and Ihnatsenka (2009b)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{mulli} \mid Li^{co}$	ST, OP	B&B
Dolgui and Ihnatsenka (2009c)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	B&B
Essafi et al. (2009)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} \mid Li^{co}$	ST, OP	GRASP
Gurevsky et al. (2009)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	GA
Dolgui, Guschinsky, Levin (2009)	DTL	$R1 \mid C^{prec}, C^{uc}, C^{exc}, SH^{mulli} \mid Li^{co}$	ST, OP	MM, G
Essafi, Delorme, Dolgui, Guschinskaya (2010)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} \mid NM$	ST, OP	MM
Essafi et al. (2010a)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} \mid Li^{co}$	ST, OP	ACO
Borisovsky et al. (2011)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST, OP	RM
Guschinskaya et al. (2011)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, LU^{i}, SH^{multi} \mid Li^{co}$	ST, OP	GRASP, GA
Borisovsky et al. (2012b)	DTL	S1 C ^{prec} , C ^{inc} , C ^{exc} , SH ^{mult} Li ^{co}	ST, OP	SP
Borisovsky et al. (2012c) Borisovsky, Delorme et al.	RTL	$S \mid C^{prec}, C^{im}, C^{exc}, SH^{mini} \mid Li^{co}$ $PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, Seq, SH^{mono} \mid NM$	ST, OP ST, OP	SP DP
(2012)	DTI	S1 Cprec Cinc Cexc S LImulti 1 :co	ST OD	мм
Forsafi et al. (2012)	DIL DTI	$S_1 C^{rec}, C^{rec}, S_H^{rec} L^{co}$ $PS_1 O Cprec Cinc Cexc A_2^C SASt SIImono T:co$	SI, OP	
Maksoud et al (2012)	RTL	$S0 + 1 \mid C^{prec} C^{inc} C^{exc} LU^{ti} SH^{multi} + I i^{co}$	ST, OP	MM
Deforme et al. (2012)	DTL	$S1 \mid C^{prec} C^{inc} C^{exc} SH^{multi} \mid Li^{co}$	ST. OP	CA
Battaïa and Dolgui (2012)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, SH^{multi} \mid Li^{co}$	ST. OP	RM
Wang and Koren (2012)	RTL	$PS0 + n \mid C^{prec} \mid NM$	ST, OP	GA
Battaïa et al. (2013)	DTL	$R1 \mid C^{prec}, C^{inc}, C^{exc}, A^{tu}, TaR^{ti}, TuR^{ti}, SH^{multi} \mid Li^{co}$	ST, OP	MM, D
Borisovsky et al. (2013)	RTL	$PS1 + 0 C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} NM$	ST, OP	GA
Bhale et al. (2014)	DTL	$S1 \mid C^{prec}, C^{inc}, C^{exc}, A^{to}, To^{ma}, Seq, TC^{ti}, Re^{ti}, Tr^{ti}, OC^{ti}, SH^{mono} \mid NonP^{ti}$	ST, TA, OP	MM
Dolgui et al. (2014)	RTL	$Snb \mid C^{prec}, To^{ma}, SH^{multi} \mid Li^{co}$	ST, OP	MM, H
Borisovsky et al. (2014)	RTL	$PS1 + 0 C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} NM$	ST, OP	CG, B&C, parallel B&C
Makssoud et al. (2014)	RTL	$S0 + 1 \mid C^{prec}, C^{inc}, C^{exc}, LU^{ti}, SH^{multi} \mid Li^{co}$	ST, OP	MM
Usman and Baki (2014)	DTL	S1 C ^{prec} , C ^{inc} , C ^{exc} , A ^{to} , To ^{ma} , Seq, TC ^{ti} , Re ^{ti} , Tr ^{ti} , OC ^{ti} , SH ^{mono} NonP ^{ti}	ST, TA, OP	MM, D, D+ACO, ACO
Battaia et al. (2014a)	DTL	$K1 \mid C^{prec}, C^{unc}, C^{exc}, TaR^{u}, TuR^{u}, SH^{multi} \mid Li^{co}$	ST, OP	MM, G
Battala et al. (2014b)	DTL	$KI \mid C^{perc}, C^{inc}, C^{exc}, LU^n, SH^{man} \mid Lt^{eo}$	ST, OP	MM, G, H
Lin et al. (2016)	KIL DTI	$FS1 + 0 C^{res}, C^{res}, C^{C}, Sd^{n}, SH^{mons} Ll^{s}, PK$ $PS1 + 0 C^{pres} Cinc C^{exc} A_{2}^{C} T_{2}^{ma} TCli OCli SJSt SHmono MM C_{1} I DD$	SI, OP	MM H
Afrin et al. (2016)	DTL	$R1 \mid C^{prec}, C^{exc}, TC^{li}, TaR^{li}, TuR^{li}, SH^{mono} \mid Li^{co}, CE$	ST, OP	MM, TS

(continued on next page)

Table 2 (continued).

Koren et al. (2017)	RTL	$PS0 + n \mid C^{prec} \mid NM$	ST, OP	MM, GA
Battaïa et al. (2017a)	FTL	Rn C ^{prec} , C ^{inc} , C ^{exc} , A ^{tu} , TC ^{ti} , TaR ^{ti} , TuR ^{ti} , SH ^{mono} Li ^{co}	ST, TA, OP	MM, H
Battaïa et al. (2017b)	FTL	$Rn \mid C^{prec}, C^{inc}, C^{exc}, A^{tu}, TC^{ti}, TaR^{ti}, TuR^{ti}, SH^{multi} \mid Li^{co}$	ST, TA, OP	MM
Xuemei et al. (2017)	DTL	$S1 \mid Bu^{al}, A^f \mid Li^{co}, PR$	ST, OP	Sim, GA-PSO
He et al. (2018)	RTL	$C1 + 0 \mid C^{prec}, SH^{mono} \mid Cyc^{ti}$	ST, OP	MM, H
Osman and Baki (2018)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, A^{to}, Ac^{C}, To^{ma}, Seq, TC^{ti}, LU^{ti}, OC^{ti}, SH^{mono} \mid NonP^{ti}, Man^{co} \mid C^{ti}, SH^{mono} \mid NonP^{ti}, Man^{co} \mid Sh^{ti}, S$	ST, TA, OP	MM, CS
Shao et al. (2019)	RTL	$PS0 + n \mid C^{prec} \mid NM$	ST, OP	MM, GA
He, Guan, Gong et al. (2019)	RTL	$C1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, SH^{mono} \mid Li^{co}$	ST, OP	MM, B&B, DB&B
He, Guan, Luo et al. (2019)	RTL	$C1 + 0 \mid C^{prec}, SH^{mono} \mid NM$	ST	MM, D
Lahrichi et al. (2019)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} \mid NM$	ST, OP	PA
Borisovsky (2020)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, SH^{mono} \mid NM$	ST, OP	GA
He et al. (2020)	RTL	$C1 + 0 \mid SH^{mono} \mid Cyc^{ti}, SI, Pro$	ST	HB&B, HABC
Shao et al. (2020)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, A^{to}, A^{f}, Bu^{al}, SH^{mono} \mid To^{co}, PR$	ST, OP	MM, MOPSO, NSGA-II, Sim
Battaïa et al. (2020)	RTL	Snb C ^{prec} , C ^{inc} , C ^{exc} , A ^{tu} , TC ^{ti} , TuR ^{ti} , Tr ^{ti} , SH ^{multi} Li ^{co}	ST, TA, OP	MM
Liu et al. (2021)	RTL	$PS1 + 0 C^{prec}, Ac^{C}, SH^{mono} Man^{co}, Rec^{co}$	ST, OP	MM, GRASP
Pirogov et al. (2021)	DTL	$S1 \mid C^{prec}, SH^{multi} \mid SR$	ST, OP	MM, H
Lahrichi, Deroussi et al. (2021)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, Seq, SH^{mono} \mid NM$	ST, OP	MH
Lahrichi, Grangeon et al. (2021)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, Ac^{C}, Sd^{st}, Seq, SH^{mono} \mid NM$	ST, OP	PA
Telemeci and glu (2022)	DTL	$S1 \mid C^{prec}, C^{exc}, SH^{multi} \mid Cyc^{ti}$	OP	MM, B&B
Battaïa et al. (2023)	RTL	Rnb C ^{prec} , C ^{inc} , C ^{exc} , A ^{tu} , TC ^{ti} , TaR ^{ti} , TuR ^{ti} , SH ^{multi} Li ^{co}	ST, TA, OP	MM, H
Cerqueus and Delorme (2023)	RTL	$PS1 + 0 C^{prec} Scal$	ST, OP	CA
Delorme et al. (2023)	RTL	$PS1 + 0 \mid C^{prec} \mid Li^{co}, En^{co}, Se^{le}$	ST, OP	MH
Borisovsky (2023)	RTL	$PS1 + 0 \mid C^{prec}, C^{inc}, C^{exc}, SH^{mono}, Sd^{st} \mid NM$	ST, OP	GA
Battaïa et al. (2024)	DTL	$S_1 \mid C^{prec}, C^{inc}, C^{exc}, TaR^{li}, TuR^{li}, SH^{mono}, A^{lu} \mid Li^{co}$	ST, TA, OP	MM, H
Delorme and Gianessi (2024)	RTL	$PS1 + 0 \mid C^{prec}, Sch \mid En^{co}$	ST, OP	MM, MH
Telemeci and Azizoğlu	DTL	$S1 \mid C^{prec}, C^{exc}, SH^{multi} \mid Cyc^{ti}$	OP	MM, B&B
Nourmohammadi et al.	DTL	$S1 \mid C^{prec}, C^{exc}, C^{iec}, C^{inc}, C^{acc}, SH^{mono}, TC^{ii}, TaR^{ii}, Sch \mid Cyc^{ii}, En^{co}$	OP	MM
Beldar et al. (2024)	DTL	$S1 \mid C^{prec}, C^{exc}, C^{inc}, C^{acc}, SH^{mono}, TC^{ii}, TaR^{ii} \mid Cyc^{ii}$	OP	Н



Fig. 6. Proportion of the type of objective functions in the literature.

as it requires the optimization of several conflicting objectives, and the solutions are typically represented as a set of trade-offs known as Pareto-optimal solutions. The low percentage of MO studies might reflect the increased complexity and computational effort required for such analyses, which can be challenging with current methodologies or computational resources. As a result, the chart highlights a notable research gap in MO optimization within TLBP and suggests possible directions for future studies to investigate more intricate, but maybe more thorough, solutions to TLBP.

Several approaches to tackling the TLBP have been suggested in the literature. To evaluate these methods, they are categorized into six groups: heuristic, metaheuristic, hybrid approach, mathematical models, exact algorithm, and simulation. From Fig. 9, it can be inferred that exact methods and mathematical models are the most frequently used approaches in TLBP literature, especially prominent in addressing DTL.

Exact methods and mathematical models offer the advantage of finding optimal solutions for small to medium-sized problems and provide valuable benchmarks for evaluating other approaches. They excel in providing insights into problem structure and properties, which can be crucial for understanding the underlying complexities of TLBP. However, these methods face significant limitations in computational tractability for large-scale problems, often becoming impractical as problem size increases. They also struggle to incorporate complex real-world constraints and have limited applicability in dynamic or uncertain environments, which are increasingly common in modern manufacturing settings.

Metaheuristic and heuristic methods are also well-represented, suggesting their utility in tackling the complex and multifaceted challenges inherent in TLBP. These approaches excel in finding good solutions quickly for large-scale problems and offer flexibility in incorporating various constraints and objectives. Their adaptability to different problem variants makes them particularly valuable in real-world applications. However, they do not guarantee optimality, which can be a significant drawback in scenarios where finding the absolute best solution is critical. Additionally, their performance can be sensitive to parameter tuning, requiring expertise to achieve optimal results.

Hybrid approaches, while less common, offer potential advantages by combining strengths of different methods. These approaches aim to leverage the benefits of multiple techniques, potentially overcoming the limitations of individual methods. However, their relative scarcity in the literature may indicate challenges in effectively integrating different methodologies or a lack of exploration in this area. The development of more sophisticated hybrid approaches could be a promising avenue for future research.

Simulation techniques show fewer connections to the various line types considered in TLBP. While they excel in modeling complex system dynamics and incorporating stochastic elements, their limited use



Fig. 7. Frequency of different objective functions in the literature.



Fig. 8. Number of objective functions in the literature of TLBP.

may indicate challenges in effectively integrating simulation with optimization approaches. Simulation methods are particularly useful for evaluating system performance under uncertainty and can incorporate complex real-world dynamics. However, they are computationally intensive and may not provide direct optimization capabilities. The results from simulation studies can also be sensitive to model assumptions and input data quality.

Fig. 10 shows the distribution of TLBP articles with an emphasis on 'Certainty' and 'Uncertainty'. Notably, the 'Certainty' portion significantly outweighs the 'Uncertainty' one, indicating that a majority of TLBP articles are based on certain or deterministic parameters. In such studies, all necessary information is assumed to be known and static, which simplifies both the modeling of problems and the development of solutions. On the other hand, the smaller but significant research component on "Uncertainty" acknowledges the complexity present in real-world production settings where variables are not always predictable. The chart shows a significant concentration of deterministic studies in the DTL category, with 47 papers compared to just one for uncertainty. To some extent, even if RTLs are supposed to provide systems able to deal with an uncertain future, the RTL category has 29 papers about certainty compared to just four for uncertainty, indicating a strong bias in favor of certainty. In the FTL category, there are three papers classified as certain and no paper as uncertain. This distribution

suggests a potential gap in the literature for developing more sophisticated models capable of managing uncertainty, which could be more relevant for dynamic and real-world manufacturing systems.

Fig. 11 indicates a clear predominance of articles focusing on strategic and operational levels of decision-making. Out of the 84 articles represented across all categories, 70 of them concern themselves with levels that are strategic and operational. This suggests that a significant majority of research in this field is dedicated to understanding and optimizing decision-making processes that are influenced by long-term strategic planning and the immediate considerations of operational management. In DTL, most articles (41 out of 48) discuss strategic and operational decision-making. RTL follows this trend, with 28 out of 33 articles also focusing on strategic and operational aspects, indicating these two levels are considered crucial in dynamic settings. FTL, although a smaller category with 3 articles, shows a pattern where strategic, tactical, and operational considerations are central in the majority of its articles (2 out of 3). Given the concentration on strategic and operational decision-making in the literature, future research could benefit from investigating the integration and impact of tactical decision-making processes, which act as a bridge between strategy and operations.

The heatmap, shown in Fig. 12, presents data for various equipmentrelated characteristics across different line types: DTL, FTL, and RTL.



Fig. 9. Frequency of the solution methods in the literature of TLBP.



Fig. 10. Distribution of TLBP articles according to parameter types.

Each cell represents the frequency of a particular characteristic within each line type. The heatmap indicates that Ac^{C} is a common characteristic across RTL, with a frequency of 15. SH^{mono} shows a high frequency in RTL, with a value of 22 and SH^{multi} shows a high frequency in DTL, with a value of 41. The analysis also reveals that Bu^{al} , $A^{f'}$, and To^{ma} are areas that have not been extensively investigated. The lack of emphasis on FTL indicates a significant research opportunity, as these parameters are critical for the optimization and flexibility of manufacturing systems.

The heatmap presented in Fig. 13 provides an insightful view of the distribution of time-related characteristics within various line types. DTL and RTL are associated with a greater diversity of these characteristics. On the other hand, FTL shows a minimal presence in the research, as indicated by the sparse frequency of articles, which may imply that time-related issues in FTL have not been extensively investigated.

From Fig. 14, the heatmap highlights a predominant focus on the parameters C^{pres} , C^{inc} , and C^{exe} , especially within DTL. DTL shows the highest frequency for these parameters, suggesting precedence, inclusion, and execution relations. In contrast, the RTL exhibits a moderate emphasis on these parameters, while the FTL presents minimal occurrences. Given this distribution, the relatively lower emphasis on sequencing and scheduling problems, denoted by *Seq* and *Sch*, may suggest a potential gap in the literature.

7. Future directions

This section focuses on responding to the third research question. The proposed classification of the available literature brings many important research paths that necessitate careful exploration and consideration. These paths are crucial in bridging the gap between research



Fig. 11. Distribution of decision-making levels across line types considered in TLBP.



Fig. 12. Distribution of equipment-related parameters across line types considered in TLBP.



Fig. 13. Distribution of time-related parameters across line types considered in TLBP.

and its practical application, improving how scholarly work affects the real world. This review study also revealed that some dimensions of TLBPs have not been extensively explored in the current body of research. In this section, the crucial importance of these often-overlooked factors is highlighted, emphasizing their pivotal role in improving the understanding and strategies for addressing TLBPs. For a methodical presentation, the insights will be divided into two distinct categories: Problem-Related Insights and Solution-Related Insights.

7.1. Problem-related insights

7.1.1. Transfer line type

Given the predominance of DTLs in the literature and the less frequent mention of FTLs and RTLs, future research should delve deeper into these less common line types. FTLs, offering more flexibility, warrant thorough exploration to understand their benefits and challenges in modern manufacturing settings. Research could focus on the



Fig. 14. Distribution of operation assignment parameters across line types considered in TLBP.

development and optimization of FTLs, assessing their applicability and efficiency across various manufacturing scenarios. A promising research avenue, driven by industrial demand, is the comparison of the efficiency of an FTL (or RTL) with several DTLs working in parallel, each dedicated to a specific product (Dolgui et al., 2021). Another potential area of study is the investigation of mixed lines, where one part functions as an FTL and another as an RTL or DTL.

7.1.2. Sustainability

Due to the recent incorporation of sustainability elements in various manufacturing environments (see Renna & Materi, 2021), integrating sustainability criteria such as energy consumption and emissions into the decision-making processes of TLBPs is crucial, particularly as industries shift towards environmentally responsible practices. The high-volume production of transfer lines significantly impacts energy use and emissions, and thus, reorienting TLBP optimization to include environmental concerns alongside traditional time and cost objectives is imperative. To the best of our knowledge, there has been limited research published in the literature that takes sustainability factors into account in TLBPs. Future research should focus on developing models prioritizing energy efficiency, emissions reduction, and overall eco-performance, balancing these with economic goals. This includes exploring trade-offs between production and sustainability objectives and employing life cycle assessments to fully understand the environmental impact of transfer line designs over their lifespan. This approach not only improves the environmental footprint but also aligns with increasing regulatory and consumer demands for greener manufacturing.

7.1.3. Uncertainty

In the domain of TLBPs, an essential direction for future research lies in exploring the uncertainties associated with cycle and completion times that go beyond conventional factors like machine breakdowns and maintenance requirements. Despite the high level of automation in manufacturing processes, operational unpredictabilities such as material quality variability, environmental conditions, intermittent tool wear, and human factors introduce significant variances in cycle and completion times. Remarkably, only 6.3% of the literature (five papers) addresses the topic of uncertainty, specifically targeting "task processing time", "processing time on machines", "demand", and "machine failures". This underscores a significant gap in the literature and a great chance for further research in this area. To address this gap, future investigations must broaden the scope of uncertainties considered in TLBP models to include those affecting cycle and completion times in more subtle yet impactful ways. This expansion necessitates a revision of traditional TLBP frameworks to integrate probabilistic or stochastic elements that capture the wider spectrum of variability in operational timings. Embracing this broader perspective on uncertainty enables the

creation of manufacturing lines that are not just efficient under ideal conditions but are also robust and flexible in the face of the varied and unpredictable challenges characteristic of real-world manufacturing environments. Stochastic programming, robust optimization, and stability radius approaches are potential decision-aid tools for generating new insights in this research direction.

7.1.4. Multi-objective optimization

Future research in TLBPs should address the current focus on singleobjective optimization, particularly the heavy emphasis on cost-related factors such as line cost, equipment cost, and the number of machines, and expand its scope to include multi-objective optimization strategies (Liu et al., 2021; Osman & Baki, 2018). This involves not only balancing cost with time-related objectives, like cycle time and non-productive time, which are crucial for throughput and overall efficiency, but also exploring the less emphasized efficiency and performance metrics like sustainability, smoothness index, and stability radius. These metrics can indirectly enhance both cost and time efficiencies and contribute to overall operational excellence. By integrating these aspects, research should aim to develop models that simultaneously address cost reduction, time management, and efficiency improvements. This multi-dimensional approach, while more complex and computationally demanding, is fundamental for achieving a balanced perspective in TLBP solutions. It requires optimizing a combination of different objectives and finding Pareto-optimal solutions that effectively balance trade-offs among these varying goals. Addressing these gaps by developing advanced methodologies and computational resources capable of handling the intricacies of multi-objective optimization, with an increased focus on efficiency and performance metrics, will significantly enhance the field, leading to more holistic and effective solutions in transfer line balancing.

7.1.5. Equipment-related consideration

The review of the existing literature reveals a notable gap at the tactical decision-making level, where intricate aspects such as A^f , A^{to} , To^{ma} , and Bu^{al} have rarely been explored. The dynamic allocation of buffers (Bu^{al}) is not only about ensuring workflow fluidity but also about maximizing the effective use of workspace and resources. The correct assignment of fixtures (A^f) is crucial for maintaining product quality and reducing cycle times by enabling quick changes between production processes. Similarly, tool assignment (A^{to}) is fundamental for operational efficiency and the minimization of machine downtime. Furthermore, the capacity of tool magazines (To^{ma}) is instrumental in ensuring that the right tools are available when needed, preventing delays, and maintaining the pace of production, which is especially crucial in FTL. Future research should pivot towards these overlooked characteristics, given their critical impact on the precision, flexibility, and uninterrupted operation of production systems at a tactical level.

7.1.6. Time-related consideration

Building upon the existing literature, it is clear that the completion time of tasks on machines in real-world settings is heavily influenced by critical time-related characteristics. Despite this, the current body of research within TLBP significantly overlooks these aspects, especially in the context of FTL. Although some time-related characteristics such as (Sd^{st}) and (OC^{ti}) have been acknowledged in specific line types, their study within FTL is markedly lacking. Parameters such as (Re^{ti}) , (LU^{ti}) , and (Tr^{ti}) are essential to accurately measure the task completion times and, thus, the overall efficiency of production systems. These elements contribute significantly to creating a realistic and holistic model of production processes, which is crucial for the optimization and enhancement of TLBPs. Future research should prioritize these overlooked influential characteristics.

7.1.7. Operation assignment consideration

The body of the TLBP literature indicates a significant research gap in the area of sequencing and scheduling problems (*Seq* and *Sch*) within the domain of TLBP across various types of lines. This gap is particularly concerning given the complexity of real-world manufacturing environments, where both the order of tasks (sequencing) and the allocation of tasks over time (scheduling) can dramatically impact the efficiency and performance of transfer lines. The scarcity of research in this area highlights a compelling opportunity for future investigation. Addressing both sequencing and scheduling challenges in TLBP could provide valuable insights into optimizing task arrangements and timing, ultimately enhancing the throughput and adaptability of these systems.

7.1.8. Industry 4.0

Due to the recent developments in Industry 4.0 emerging in various manufacturing settings (Facchini et al., 2023; Hosseinzadeh et al., 2023; Mahmoodi et al., 2024; Vlachos et al., 2022), incorporating Industry 4.0 technologies, such as digital twin, Artificial Intelligence (AI), and the Internet of Things (IoT), into TLBPs presents significant research opportunities.

Industry 4.0 technologies, particularly IoT sensors, offer unprecedented capabilities for collecting accurate data from machines within transfer lines. These sensors continuously monitor various parameters, including machine performance, downtime events, and energy usage across different operational modes. This granular data collection enables the creation of highly accurate digital twins (virtual replicas of the physical transfer line). Digital twins provide new possibilities for TLBP simulation and optimization. These virtual models allow rapid testing of various line balancing strategies without disrupting actual production. By incorporating real-time data on machine downtimes and energy consumption, digital twins can simulate more realistic scenarios, enabling manufacturers to evaluate multiple configurations and select the most efficient and energy-optimized transfer line setup.

With leveraging historical data from IoT sensors and applying machine learning algorithms, it becomes possible to accurately forecast potential equipment failures across workstations. This capability is especially valuable in transfer lines, where the malfunction of a single station can disrupt the entire production flow. Additionally, with the use of AI techniques it becomes possible to evaluate energy usage patterns of machines in various operational modes, providing valuable insights for optimizing energy consumption. These insights enable the development of TLBP solutions that integrate energy efficiency with traditional production metrics. The incorporation of AI-driven predictive maintenance and energy optimization ensures a holistic approach to line balancing, improving production efficiency, maintenance scheduling, and energy management across the entire transfer line.

Finally, as stated by Cerqueus and Delorme (2023) there is a clear relationship between Industry 4.0 and RTL since several technologies (e.g. cyber–physical systems, collaborative robots, additive manufacturing) can act as enablers for the development of RTL, and should be increasingly considered in research works.

7.2. Solution-related insights

Hybrid approaches, which combine elements from different methods such as heuristics, exact algorithms, metaheuristics, and simulations, could offer a more comprehensive and effective way of addressing the complexities of TLBPs (He et al., 2020; Osman & Baki, 2014). They might be particularly suited to scenarios that involve a high degree of uncertainty or where the balancing problem presents unique challenges that cannot be adequately addressed by a single method. Two of the promising solution approaches that future research could focus on developing and testing their applicability and effectiveness in various TLBP scenarios are simulation-based optimization and the integration of machine learning into meta-heuristics.

Despite the advantages of simulation models in capturing realworld scenarios and uncertainties, they have rarely been used in the context of TLBP. Simulation offers a unique perspective, particularly for large-scale systems, which often pose challenges for purely mathematical models. By simulating the dynamics of transfer line operations, researchers can gain valuable insights into how these systems perform in actual scenarios and under various conditions. This approach allows for a more comprehensive understanding of the intricate interactions between machines, workstations, and the production environment (Shao et al., 2020, 2022). Future research in TLBPs can focus on the development of efficient integration schemes that combine optimization and simulation techniques, forming a simulation-based optimization approach. This integration is crucial for streamlining the design process of transfer lines. Through a seamless integration of optimization algorithms with simulation models, researchers can create a closed-loop design framework. This framework enables the optimization algorithms to make decisions based on the feedback from the simulation while taking the production and equipment uncertainty into consideration.

In recent years, there has also been a growing interest in integrating machine learning techniques into meta-heuristics to solve combinatorial optimization problems (Karimi-Mamaghan et al., 2022). However, this has not yet been applied to the TLBP. This integration could guide meta-heuristics towards more efficient, effective, and robust searches, improving their performance in terms of solution quality, convergence rate, and robustness. Machine learning can help in understanding the search space better, predicting the performance of solutions, or even guiding the search process based on historical data. Such an approach could significantly enhance the capabilities of meta-heuristics in handling the complexity and dynamic nature of TLBPs.

8. Conclusion

This systematic review study examines TLBP literature, proposing a tuple notation classification framework that includes system configuration, specific problem characteristics, and objective functions. It draws attention to the complexities of the TLBP, revealing a prominent focus on DTLs while identifying significant gaps in the study of FTLs. The study emphasizes the importance of incorporating sustainability into TLBP techniques, which has been largely disregarded in the previous literature. It also highlights the scarcity of multi-objective optimization methods in TLBP investigations, suggesting this is an important field for future research. Furthermore, the study of hybrid techniques, the management of uncertainty, and the equipment-related, time-related, operation assignment considerations are highlighted, emphasizing their significance in improving the adaptability and efficiency of manufacturing lines. This study represents a critical turning point in TLBP research by emphasizing the necessity for thorough, long-lasting, and effective manufacturing solutions.

CRediT authorship contribution statement

Pedram Beldar: Writing - review & editing, Writing original draft, Visualization, Validation, Software, Project administration, Methodology, Investigation, Formal analysis, Data curation, Conceptualization. Masood Fathi: Writing - review & editing, Writing - original draft, Validation, Supervision, Resources, Project administration, Methodology, Investigation, Funding acquisition, Formal analysis, Conceptualization. Amir Nourmohammadi: Writing - review & editing, Writing - original draft, Validation, Supervision, Project administration, Methodology, Investigation, Formal analysis, Conceptualization. Xavier Delorme: Writing - review & editing, Validation, Supervision, Methodology, Investigation, Formal analysis, Conceptualization. Olga Battaïa: Writing - review & editing, Validation, Supervision, Methodology, Investigation, Formal analysis, Conceptualization. Alexandre Dolgui: Writing - review & editing, Validation, Supervision, Methodology, Investigation, Conceptualization.

Acknowledgments

This study is partly supported by the Knowledge Foundation (KKS) and Sweden's Innovation Agency through the ACCURATE 4.0 project (Grant No. 20200181) and the PREFER project, respectively. It is also partly supported by the project ANR-21-CE10-0019 "RECON-FIDURABLE" and by the project Carnot M.I.N.E.S. "Industrie responsible".

Data availability

Data will be made available on request.

References

- Afrin, K., Iquebal, A. S., Kumar, S. K., Tiwari, M., Benyoucef, L., & Dolgui, A. (2016). Towards green automated production line with rotary transfer and turrets: a multiobjective approach using a binary scatter tabu search procedure. *International Journal of Computer Integrated Manufacturing*, 29(7), 768–785.
- Battaïa, O., & Dolgui, A. (2012). Reduction approaches for a generalized line balancing problem. Computers & Operations Research, 39(10), 2337–2345.
- Battaïa, O., & Dolgui, A. (2013). A taxonomy of line balancing problems and their solutionapproaches. International Journal of Production Economics, 142(2), 259–277.
- Battaïa, O., & Dolgui, A. (2022). Hybridizations in line balancing problems: A comprehensive review on new trends and formulations. *International Journal of Production Economics*, 250, Article 108673.
- Battaïa, O., Dolgui, A., & Guschinsky, N. (2017a). Decision support for design of reconfigurable rotary machining systems for family part production. *International Journal of Production Research*, 55(5), 1368–1385.
- Battaïa, O., Dolgui, A., & Guschinsky, N. (2017b). Integrated process planning and system configuration for mixed-model machining on rotary transfer machine. *International Journal of Computer Integrated Manufacturing*, 30(9), 910–925.
- Battaïa, O., Dolgui, A., & Guschinsky, N. (2020). Optimal cost design of flow lines with reconfigurable machines for batch production. *International Journal of Production Research*, 58(10), 2937–2952.
- Battaïa, O., Dolgui, A., & Guschinsky, N. (2023). MIP-based heuristics for combinatorial design of reconfigurable rotary transfer machines for production of multiple parts. *International Journal of Production Economics*, 262, Article 108904.
- Battaïa, O., Dolgui, A., & Guschinsky, N. (2024). An exact method for machining lines design with equipment selection and line balancing. *International Journal of Production Research*, 62(1–2), 71–91.
- Battaïa, O., Dolgui, A., Guschinsky, N., & Levin, G. (2005). A combined heuristic approach for optimization of a class of machining lines. vol. 2, In 2005 IEEE international conference on automation science and engineering (pp. 154–159).
- Battaïa, O., Dolgui, A., Guschinsky, N., & Levin, G. (2013). Parallel machining of multiple parts on rotary transfer machines with turrets. *IFAC Proceedings Volumes*, 46(9), 1477–1482.
- Battaïa, O., Dolgui, A., Guschinsky, N., & Levin, G. (2014a). Combinatorial techniques to optimally customize an automated production line with rotary transfer and turrets. *IIE Transactions*, 46(9), 867–879.
- Battaïa, O., Dolgui, A., Guschinsky, N., & Levin, G. (2014b). Integrated configurable equipment selection and line balancing for mass production with serial-parallel machining systems. *Engineering Optimization*, 46(10), 1369–1388.

- Beldar, P., Nourmohammadi, A., Fathi, M., & Ng, A. H. (2024). A heuristic approach for flexible transfer line balancing problem. *Proceedia CIRP*, 130, 1144–1149.
- Belmokhtar, S., Dolgui, A., Delorme, X., & Ignatenko, I. (2006). Optimizing modular machining line design problem. *IFAC Proceedings Volumes*, 39(3), 443–448.
- Bhale, S., Baki, M. F., & Azab, A. (2014). Grouping and sequencing of machining operations for high volume transfer lines. *Proceedia CIRP*, 17, 76–81.
- Borisovsky, P. (2020). Genetic algorithm for one machining line balancing problem with setup times. In 2020 dynamics of systems, mechanisms and machines (dynamics) (pp. 1–5).
- Borisovsky, P. (2023). Parallel evolutionary algorithms for the reconfigurable transfer line balancing problem. Yugoslav Journal of Operations Research, 34(1), 93–107.

Borisovsky, P. A., Delorme, X., & Dolgui, A. (2012). Balancing reconfigurable machining lines by means of set partitioning model. *IFAC Proceedings Volumes*, 45(6), 426–431.

- Borisovsky, P. A., Delorme, X., & Dolgui, A. (2013). Genetic algorithm for balancing reconfigurable machining lines. Computers & Industrial Engineering, 66(3), 541–547.
- Borisovsky, P. A., Delorme, X., & Dolgui, A. (2014). Balancing reconfigurable machining lines via a set partitioning model. *International Journal of Production Research*, 52(13), 4026–4036.
- Borisovsky, P., Dolgui, A., & Kovalev, S. (2011). A new model for equipment selection and transfer line design problem. *IFAC Proceedings Volumes*, 44(1), 3962–3967.
- Borisovsky, P., Dolgui, A., & Kovalev, S. (2012b). Algorithms and implementation of a set partitioning approach for modular machining line design. *Computers & Operations Research*, 39(12), 3147–3155.
- Borisovsky, P., Dolgui, A., & Kovalev, S. (2012c). Modelling transfer line design problem via a set partitioning. *Optimization Letters*, 6(5), 915–926.
- Boysen, N., Fliedner, M., & Scholl, A. (2007). A classification of assembly line balancing problems. European Journal of Operational Research, 183(2), 674–693.
- Boysen, N., Schulze, P., & Scholl, A. (2022). Assembly line balancing: What happened in the last fifteen years? European Journal of Operational Research, 301(3), 797–814.
- Bratcu, A., Dolgui, A., & Belmokhtar, S. (2005). Reconfigurable transfer lines cost optimization - a linear programming approach. vol. 1, In 2005 IEEE conference on emerging technologies and factory automation (p. 8). pp.632.
- Bratcu, A., Makdessian, L., & Dolgui, A. (2003). Minimisation of equipment cost for transfer lines with blocks of parallel tasks. In *Proceedings of the IEEE international* symposium onAssembly and task planning, 2003 (pp. 109–114).
- Brucker, P., Drexl, A., Möhring, R., Neumann, K., & Pesch, E. (1999). Resourceconstrained project scheduling: Notation, classification, models, and methods. *European Journal of Operational Research*, 112(1), 3–41.
- Cerqueus, A., & Delorme, X. (2023). Evaluating the scalability of reconfigurable manufacturing systems at the design phase. *International Journal of Production Research*, 61(23), 8080–8093.
- Delorme, X., Cerqueus, A., Gianessi, P., & Lamy, D. (2023). RMS balancing and planning under uncertain demand and energy cost considerations. *International Journal of Production Economics*, 261, Article 108873.
- Delorme, X., Dolgui, A., & Kovalyov, M. Y. (2012). Combinatorial design of a minimum cost transfer line. *Omega*, 40(1), 31–41.
- Delorme, X., & Gianessi, P. (2024). Line balancing and task scheduling to minimise power peak of reconfigurable manufacturing systems. *International Journal of Production Research*, 62(14), 5061–5086.
- Delorme, X., Malyutin, S., & Dolgui, A. (2016). A multi-objective approach for design of reconfigurable transfer lines. *IFAC-PapersOnLine*, 49(12), 509–514.
- Deniz, N., & Ozcelik, F. (2019). An extended review on disassembly line balancing with bibliometric and social network and future study realization analysis. *Journal* of Cleaner Production, 225, 697–715.
- Dolgui, A., Eremeev, A., & Guschinskaya, O. (2010). MIP-based GRASP and genetic algorithm for balancing transfer lines. In *Matheuristics: hybridizing metaheuristics and mathematical programming* (pp. 189–208). Boston, MA: Springer US.
- Dolgui, A., Finel, B., Guschinskaya, O., Guschinsky, N., Levin, G., & Vernadat, F. B. (2006a). Balancing large-scale machining lines with multi-spindle heads using decomposition. *International Journal of Production Research*, 44(18–19), 4105–4120.
- Dolgui, A., Finel, B., Guschinsky, N., Levin, G. M., & Vernadat, F. B. (2003). Some optimization approaches for transfer lines with blocks of operations. *IFAC Proceedings Volumes*, 36(3), 261–266.
- Dolgui, A., Finel, B., Guschinsky, N., Levin, G. M., & Vernadat, F. B. (2006b). MIP approach to balancing transfer lines with blocks of parallel operations. *IIE Transactions*, 38(10), 869–882.
- Dolgui, A., Finel, B., Vernadat, F. B., Guschinsky, N., & Levin, G. M. (2005). A heuristic approach for transfer lines balancing. *Journal of Intelligent Manufacturing*, 16, 159–172.
- Dolgui, A., Guschinskaya, O., Guschinsky, N., & Levin, G. M. (2009). Optimisation of machining lines composed of unit-built machines. *IFAC Proceedings Volumes*, 42(4), 1205–1210.
- Dolgui, A., Guschinsky, N., & Levin, G. M. (1999). On problem of optimal design of transfer lines with parallel and sequential operations. vol. 1, In 1999 7th IEEE international conference on emerging technologies and factory automation. proceedings ETFA '99 (cat. nOlga99TH8467) (pp. 329–334).
- Dolgui, A. B., Guschinsky, N., & Levin, G. M. (2000). Decomposition methods to optimise transfer line with parallel and sequential machining. *IFAC Proceedings Volumes*, 33(17), 983–988.

- Dolgui, A., Guschinsky, N., & Levin, G. (2005). Transfer line balancing by a combined approach. IFAC Proceedings Volumes, 38(1), 277–282.
- Dolgui, A., Guschinsky, N., & Levin, G. M. (2006c). A decomposition method for transfer line life cycle cost optimisation. *Journal of Mathematical Modelling and Algorithms*, 5(2), 215–238.
- Dolgui, A., Guschinsky, N., & Levin, G. (2006d). A special case of transfer lines balancing by graph approach. European Journal of Operational Research, 168(3), 732–746.
- Dolgui, A., Guschinsky, N., & Levin, G. (2008). Exact and heuristic algorithms for balancing transfer lines when a set of available spindle heads is given. *International Transactions in Operational Research*, 15(3), 339–357.
- Dolgui, A., Guschinsky, N., & Levin, G. M. (2009). Graph approach for optimal design of transfer machine with rotary table. *International Journal of Production Research*, 47(2), 321–341.
- Dolgui, A., Guschinsky, N., & Levin, G. (2012). Enhanced mixed integer programming model for a transfer line design problem. *Computers & Industrial Engineering*, 62(2), 570–578.
- Dolgui, A., Guschinsky, N., Levin, G. M., & Harrath, Y. (2000). Optimal design of a class of transfer lines with blocks of parallel operations. *IFAC Proceedings Volumes*, 33(20), 37–42.
- Dolgui, A., Guschinsky, N., Levin, G., Louly, M.-A., & Belmokhtar, S. (2004). Balancing of transfer lines with simultaneously activated spindles. *IFAC Proceedings Volumes*, 37(4), 45–50.
- Dolgui, A., Guschinsky, N., Levin, G. M., & Proth, J.-M. (2008). Optimisation of multiposition machines and transfer lines. *European Journal of Operational Research*, 185(3), 1375–1389.
- Dolgui, A., Hashemi-Petroodi, S. E., Kovalev, S., & Kovalyov, M. Y. (2021). Profitability of a multi-model manufacturing line versus multiple dedicated lines. *International Journal of Production Economics*, 236, Article 108113.
- Dolgui, A., & Ihnatsenka, I. (2005). Machining lines with multi-spindle workstations: a new optimization problem. vol. 1, In 2005 IEEE conference on emerging technologies and factory automation (pp. 8 pp.-474).
- Dolgui, A., & Ihnatsenka, I. (2009a). An approach to transfer line balancing via a special set partitioning problem. *IFAC Proceedings Volumes*, 42(4), 750–755.
- Dolgui, A., & Ihnatsenka, I. (2009b). Balancing modular transfer lines with serialparallel activation of spindle heads at stations. *Discrete Applied Mathematics*, 157(1), 68–89.
- Dolgui, A., & Ihnatsenka, I. (2009c). Branch and bound algorithm for a transfer line design problem: Stations with sequentially activated multi-spindle heads. *European Journal of Operational Research*, 197(3), 1119–1132.
- Dolgui, A., Kovalev, S., Kovalyov, M. Y., Nossack, J., & Pesch, E. (2014). Minimizing setup costs in a transfer line design problem with sequential operation processing. *International Journal of Production Economics*, 151, 186–194.
- Dolgui, A., Levin, G., & Rozin, B. (2020). Optimisation of the aggregation and execution rates for intersecting operation sets: an example of machining process design. *International Journal of Production Research*, 58(9), 2658–2676.
- Eren Özceylan, A. G., & Gupta, S. M. (2019). Disassembly line balancing problem: a review of the state of the art and future directions. *International Journal of Production Research*, 57(15–16), 4805–4827.
- Essafi, M., Delorme, X., & Dolgui, A. (2009). A GRASP heuristic for sequence-dependent transfer line balancing problem. *IFAC Proceedings Volumes*, 42(4), 762–767, 13th IFAC Symposium on Information Control Problems in Manufacturing.
- Essafi, M., Delorme, X., & Dolgui, A. (2010a). Balancing lines with CNC machines: A multi-start ant based heuristic. CIRP Journal of Manufacturing Science and Technology, 2(3), 176–182.
- Essafi, M., Delorme, X., & Dolgui, A. (2010b). New trends and challenges in transfer line balancing. In Proceedings of APMS 2010 - international conference on advances in production management systems.
- Essafi, M., Delorme, X., & Dolgui, A. (2012). A reactive GRASP and path relinking for balancing reconfigurable transfer lines. *International Journal of Production Research*, 50(18), 5213–5238.
- Essafi, M., Delorme, X., Dolgui, A., & Guschinskaya, O. (2010). A MIP approach for balancing transfer line with complex industrial constraints. *Computers & Industrial Engineering*, 58(3), 393–400.
- Facchini, F., Mossa, G., Sassanelli, C., & Digiesi, S. (2023). IoT-based milk-run routing for manufacturing system: an application case in an automotive company. *International Journal of Production Research*, 1–20.
- Finel, B., Dolgui, A., & Vernadat, F. B. (2008). A random search and backtracking procedure for transfer line balancing. *International Journal of Computer Integrated Manufacturing*, 21(4), 376–387.
- Graham, R., Lawler, E. L., Lenstra, J. K., & Kan, A. H. R. (1979). Optimization and approximation in deterministic sequencing and scheduling: a survey. vol. 5, In Discrete optimization II (pp. 287–326). Elsevier.
- Gurevsky, E., Battaïa, O., Eremeev, A., & Dolgui, A. (2009). Balancing machining transfer lines using genetic algorithms. (pp. 1832–1837).
- Guschinskaya, O., & Dolgui, A. (2006). A comparative evaluation of exact and heuristic methods for transfer line balancing problem. *IFAC Proceedings Volumes*, 39(3), 413–418.
- Guschinskaya, O., & Dolgui, A. (2007). Balancing transfer lines with multi-spindle machines using GRASP. IFAC Proceedings Volumes, 40(18), 511–516.

- Guschinskaya, O., & Dolgui, A. (2008). A transfer line balancing problem by heuristic methods: Industrial case studies. *Decision Making in Manufacturing and Services*, 2(1–2), 33–46.
- Guschinskaya, O., & Dolgui, A. (2009). Comparison of exact and heuristic methods for a transfer line balancing problem. *International Journal of Production Economics*, 120(2), 276–286.
- Guschinskaya, O., Dolgui, A., Guschinsky, N., & Levin, G. M. (2007). New reduction methods for the transfer line balancing problem. *IFAC Proceedings Volumes*, 40(3), 69–74.
- Guschinskaya, O., Dolgui, A., Guschinsky, N., & Levin, G. M. (2008). A heuristic multistart decomposition approach for optimal design of serial machining lines. *European Journal of Operational Research*, 189(3), 902–913.
- Guschinskaya, O., Gurevsky, E., Dolgui, A., & Eremeev, A. (2011). Metaheuristic approaches for the design of machining lines. *International Journal of Advanced Manufacturing Technology*, 55(1), 11–22.
- He, C., Guan, Z., Gong, Y., Wang, C., & Yue, L. (2019). Automated flexible transfer line design problem: Sequential and reconfigurable stages with parallel machining cells. *Journal of Manufacturing Systems*, 52, 157–171.
- He, C., Guan, Z., Luo, D., Fang, W., & Ullah, S. (2019). Decomposition heuristic for parallel-machine transfer line design with dual-uncertainties-based chance constraints. *Engineering Optimization*, 51(2), 199–216.
- He, C., Guan, Z., Xu, G., Yue, L., & Ullah, S. (2020). Scenario-based robust dominance criteria for multi-objective automated flexible transfer line balancing problem under uncertainty. *International Journal of Production Research*, 58(2), 467–486.
- He, C., Guan, Z., Yue, L., & Ullah, S. (2018). Set-partitioning-based heuristic for balancing and configuration of automated flexible machining line. *International Journal of Production Research*, 56(9), 3152–3172.
- Hosseinzadeh, A., Chen, F. F., Shahin, M., & Bouzary, H. (2023). A predictive maintenance approach in manufacturing systems via AI-based early failure detection. *Manufacturing Letters*, 35, 1179–1186.
- Karimi-Mamaghan, M., Mohammadi, M., Meyer, P., Karimi-Mamaghan, A. M., & Talbi, E.-G. (2022). Machine learning at the service of meta-heuristics for solving combinatorial optimization problems: A state-of-the-art. *European Journal of Operational Research*, 296(2), 393–422.
- Koren, Y., Wang, W., & Gu, X. (2017). Value creation through design for scalability of reconfigurable manufacturing systems. *International Journal of Production Research*, 55(5), 1227–1242.
- Lahrichi, Y., Deroussi, L., Grangeon, N., & Norre, S. (2019). A polynomial algorithm for balancing a sequence of operations in reconfigurable transfer lines. *IFAC-PapersOnLine*, 52(13), 981–986.
- Lahrichi, Y., Deroussi, L., Grangeon, N., & Norre, S. (2021). A balance-first sequencelast algorithm to design RMS: a matheuristic with performance guaranty to balance reconfigurable manufacturing systems. *Journal of Heuristics*, 27(1), 107–132.
- Lahrichi, Y., Grangeon, N., Deroussi, L., & Norre, S. (2021). A new split-based hybrid metaheuristic for the reconfigurable transfer line balancing problem. *International Journal of Production Research*, 59(4), 1127–1144.
- Liu, X., Chen, J., & Li, A. (2021). Optimisation of line configuration and balancing for reconfigurable transfer lines considering demand uncertainty. *International Journal* of Production Research, 59(2), 444–466.
- Liu, X., Li, A., & Chen, Z. (2016). Optimization of line configuration and balancing for flexible machining lines. *Chinese Journal of Mechanical Engineering*, 29(3), 579–587.
- Mahmoodi, E., Fathi, M., Tavana, M., Ghobakhloo, M., & Ng, A. H. (2024). Data-driven simulation-based decision support system for resource allocation in industry 4.0 and smart manufacturing. *Journal of Manufacturing Systems*, 72, 287–307.
- Makssoud, F., Battaïa, O., & Dolgui, A. (2012). A mathematical model for a reconfiguration problem of transfer machining lines. *IFAC Proceedings Volumes*, 45(6), 438–443.
- Makssoud, F., Battaïa, O., & Dolgui, A. (2014). An exact optimization approach for a transfer line reconfiguration problem. *International Journal of Advanced Manufacturing Technology*, 72(5), 717–727.
- Masood, S. (2006). Line balancing and simulation of an automated production transfer line. Assembly Automation, 26(1), 69–74.
- Moher, D., Liberati, A., Tetzlaff, J., Altman, D. G., & PRISMA Group*, t. (2009). Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. Annals of Internal Medicine, 151(4), 264–269.
- Nourmohammadi, A., Beldar, P., Fathi, M., & Mahmoodi, E. (2024). Balancing and scheduling of sustainable flexible transfer lines. *IFAC-PapersOnLine*, 58(19), 664–669.
- Osman, H., & Baki, M. F. (2014). Balancing transfer lines using benders decomposition and ant colony optimisation techniques. *International Journal of Production Research*, 52(5), 1334–1350.
- Osman, H., & Baki, M. F. (2018). A cuckoo search algorithm to solve transfer line balancing problems with different cutting conditions. *IEEE Transactions on Engineering Management*, 65(3), 505–518.
- Pirogov, A., Gurevsky, E., Rossi, A., & Dolgui, A. (2021). Robust balancing of transfer lines with blocks of uncertain parallel tasks under fixed cycle time and space restrictions. *European Journal of Operational Research*, 290(3), 946–955.
- Renna, P., & Materi, S. (2021). A literature review of energy efficiency and sustainability in manufacturing systems. *Applied Sciences*, 11(16).

- Salveson, M. E. (1955). The assembly-line balancing problem. Transactions of the American Society of Mechanical Engineers, 77(6), 939–947.
- Shao, H., Li, A., Xu, L., & Moroni, G. (2019). Scalability in manufacturing systems: a hybridized GA approach. *Journal of Intelligent Manufacturing*, 30, 1859–1879.
- Shao, H., Moroni, G., Li, A., Liu, X., & Xu, L. (2020). Simultaneously solving the transfer line balancing and buffer allocation problems with a multi-objective approach. *Journal of Manufacturing Systems*, 57, 254–273.
- Shao, H., Moroni, G., Li, A., & Xu, L. (2022). Heuristic approach for a combined transfer line balancing and buffer allocation problem considering uncertain demand. *Applied Sciences*, 12(12).
- Telemeci, Y. E., & Azizoğlu, M. (2024). Type-II transfer line balancing problem a branch and bound approach. *Computers & Industrial Engineering*, 198, Article 110689.
- Telemeci, Y. E., & glu, M. A. (2022). An exact solution approach to the type-II transfer line balancing problem. *IFAC-PapersOnLine*, 55(10), 442–447.
- Vlachos, I., Pascazzi, R. M., Ntotis, M., Spanaki, K., Despoudi, S., & Repoussis, P. (2022). Smart and flexible manufacturing systems using autonomous guided vehicles (AGVs) and the internet of things (IoT). *International Journal of Production Research*, 1–22.
- Wang, W., & Koren, Y. (2012). Scalability planning for reconfigurable manufacturing systems. Journal of Manufacturing Systems, 31(2), 83–91.
- Wohlin, C. (2014). Guidelines for snowballing in systematic literature studies and a replication in software engineering. In *Proceedings of the international conference on evaluation and assessment in software engineering* (pp. 321-330).
- Xuemei, L., Huan, S., Rui, Z., Yongqi, J., & Aiping, L. (2017). Collaborative optimization of transfer line balancing and buffer allocation based on polychromatic set. *Procedia CIRP*, 63, 213–218.